Author's response to reviews

Title: Phase I-II study of everolimus and low-dose oral cyclophosphamide in patients with metastatic renal cell cancer

Authors:

Charlotte M Huijts (c.huijts@vumc.nl)
Saskia J Santegoets (s.santegoets@vumc.nl)
Alfons J van den Eertwegh (vandeneertwegh@vumc.nl)
Laura S Pijpers (l.pijpers@vumc.nl)
John B Haanen (j.haanen@nki.nl)
Tanja D de Gruijl (td.degruijl@vumc.nl)
Henk M Verheul (h.verheul@vumc.nl)
Hans J van der Vliet (jj.vandervliet@vumc.nl)

Version: 2 Date: 28 November 2011

Author's response to reviews:

Editorial Board
BMC Cancer

Amsterdam, November 28th, 2011

Dear Editor,

Thank you for accepting our manuscript entitled “Phase I-II study of everolimus and low-dose oral cyclophosphamide in patients with metastatic renal cell cancer” for publication in BMC cancer.

Upon request by the editorial production team, we made the following changes:

- The list of authors in the manuscript has been written exactly as they are in the submission system
- All of the additional files have been removed
- Authors’ contributions have been formatted into a paragraph
- The figure has been cropped as closely as possible
- The document has been re-checked for typography and formatting errors.

Yours sincerely,

C.M. Huijts, MD
H.J. van der Vliet, MD, PhD
Department of Medical Oncology
VU University medical center
De Boelelaan 1117
1081 HV, Amsterdam, The Netherlands
Email: c.huijts@vumc.nl, jj.vandervliet@vumc.nl